

The prevalence of scabies, pyoderma and other communicable dermatoses in the Bijagos Archipelago, Guinea-Bissau

Michael Marks 1,2°*, Thomas Sammut 1°, Marito Gomes Cabral 3, Eunice Teixeira da Silva 3, Adriana Goncalves 1, Amabelia Rodrigues 4, Cristóvão Manjuba 5, Jose Nakutum 3, Janete Ca 3, Umberto D'Alessandro 6, Jane Achan 6, James Logan 7, Robin Bailey 1,2, David Mabey 1,2, Anna Last 1,2, Stephen L. Walker 1,2,8

- 1 Clinical Research Department, Faculty of Infectious and Tropical Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Hospital for Tropical Diseases, University College London Hospital NHS Foundation Trust, London, United Kingdom, 3 Region Sanitaria Bolama-Bijagós, Bubaque, Guinea Bissau, 4 Bandim Health Project, Bissau, Guinea Bissau, 5 Ministry of Public Health, Bissau, Guinea Bissau, 6 MRC The Gambia at the London School of Hygiene & Tropical Medicine, Fajara, Gambia, 7 Department of Disease Control, Faculty of Infectious and Tropical Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom, 8 Department of Dermatology, University College London Hospitals NHS Foundation Trust, London, United Kingdom
- These authors contributed equally to this work.
- * Michael.marks@lshtmac.uk



G OPEN ACCESS

Citation: Marks M, Sammut T, Cabral MG, Teixeira da Silva E, Goncalves A, Rodrigues A, et al. (2019) The prevalence of scabies, pyoderma and other communicable dermatoses in the Bijagos Archipelago, Guinea-Bissau. PLoS Negl Trop Dis 13(11): e0007820. https://doi.org/10.1371/journal.pntd.0007820

Editor: Darren J. Gray, Australian National University, AUSTRALIA

Received: June 18, 2019

Accepted: October 1, 2019

Published: November 18, 2019

Copyright: © 2019 Marks et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This work was supported by an MRC Global Challenges Research Foundation Award [grant number MR/P023843/1]. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Abstract

Introduction

Skin diseases represent a significant public health problem in most low and middle income settings. Nevertheless, there is a relative paucity of high-quality epidemiological data on the prevalence of these conditions.

Materials/Methods

We conducted two cross-sectional population-based skin-surveys of children (6 months to 9 years old) in the Bijagós Archipelago of Guinea-Bissau during the dry season (February-March 2018) and the wet season (June-July 2018). Following a period of training, a nurse performed a standardised examination for communicable dermatoses for each participant. We calculated the prevalence of each skin condition and investigated demographic associations.

Results

1062 children were enrolled in the dry season survey of whom 318 (29.9%) had at least one skin diseases. The most common diagnosis was tinea capitis (154/1062, 14.5% - 95% CI 12.5–16.8%) followed by tinea corporis (84/1062, 7.9% - 95% CI 6.4–9.7%), pyoderma (82/1062, 7.7% - 95% CI 6.2–9.5%) and scabies (56/1062. 5.2% - 95%CI 4.0–6.8%). 320 children were enrolled in the wet season survey of whom 121 (37.8%) had at least one skin problem. Tinea capitis remained the most common diagnosis (79/320, 24.7% - 95% CI



Competing interests: The authors have declared that no competing interests exist.

20.1–29.9%), followed by pyoderma (38/320, 11.9% - 95% CI 8.6–16.1%), tinea corporis (23/320, 7.2% - 95% 4.7–10.7%) and scabies (6/320, 1.9% - 95% CI 0.8–4.2%).

Conclusions

Our study, which utilised robust population-based cluster random sampling methodology, demonstrates the substantial disease burden caused by common communicable dermatoses in this setting. Given these findings, there is a need to consider common dermatoses as part of Universal Health Coverage to deliver 'skin-health for all'.

Author summary

Skin conditions are very common in many low and middle income settings but there have been relatively few surveys of skin disease conducted using the best epidemiological approaches. We performed two cross-sectional population-based skin-surveys of children in the Bijagós Archipelago, Guinea-Bissau, using gold-standard sampling methodologies. Skin conditions were extremely common and almost 30% of children had at least one common, infectious skin condition in both surveys. Fungal skin and scalp infections were the most common conditions followed by bacterial skin infections. Our survey demonstrates that common, easily treatable skin conditions are responsible for a high burden of disease in this population. Skin-health should be considered a key component of the Universal Health Coverage agenda.

Introduction

The prevalence of skin disease in many low and middle income countries is high and both communicable and non-communicable skin diseases represent a significant public health problem in these settings [1]. Skin diseases are amongst the leading causes of disability adjusted life years[1]. The World Health Organization (WHO) has established programmes for Neglected Tropical Diseases (NTDs) including several severe, stigmatising skin diseases such as Buruli ulcer, leprosy and leishmaniasis. However, more common dermatoses are responsible for a significant burden of disease, in particular fungal skin infections, scabies and pyoderma which are all amongst the 50 most prevalent diseases worldwide. Common skin infections and infestations, such as scabies, are associated with significant impairment of health-related quality of life.[2].

In view of the morbidity associated with infestation, and its association with pyoderma, scabies has recently been adopted by WHO as an NTD. The majority of high quality surveys for scabies have been conducted in the Pacific or Latin America[3] with fewer data available from Africa. Data suggest that regular mass drug administration (MDA) conducted for the control of onchocerciasis or lymphatic filariasis may be responsible for a reduced prevalence of scabies in Africa but this is not a consistent finding[4,5]. Furthermore large outbreaks of scabies have been reported in Africa in recent years including an outbreak in Ethiopia involving more than one million individuals[6].

Despite the high burden of morbidity associated with these diseases there is a relative paucity of high quality epidemiological data on their prevalence. The majority of studies have used epidemiological methods which may be prone to bias such as school based surveys or screening individuals who report a skin problem[2,7,8]. As skin diseases may be associated with



stigma and reduced school attendance these methods may misrepresent the true burden of morbidity due to these more common skin conditions in communities.

Since 2011 we have been conducting research on NTDs and other conditions of public health concern in the Bijagos Archipelago of Guinea-Bissau [9]. In 2018 we conducted a skin-survey of children to provide population level estimates of the prevalence of common infectious dermatoses in this population.

Methods

Setting

The Republic of Guinea-Bissau is located in coastal West Africa and has an estimated population of 1,792,338 with a median age of 20 years. Gross Domestic Product per capita is estimated at 1,800 USD which is among the lowest income in the world. The Bijagos Archipelago is a collection of 18 main islands and other small uninhabited islands. Bubaque is the major island of the archipelago. Health care on Bubaque is provided by a single small hospital facility alongside community healthcare provided by nursing posts and community health volunteers. The Ministry of Health (MoH) has previously conducted Mass Drug Administration (MDA) campaigns as part of programmes for the elimination of NTDs in the archipelago including azithromycin MDA for trachoma (2013–2016) and ivermectin and albendazole for lymphatic filariasis (2017–2018).

Ethical approval was obtained from the Research Ethics Committee of the London School of Hygiene and Tropical Medicine and the Comite Nacional de Etica na Saude of Guinea-Bissau. Permission was given from village leaders for the surveys to take place. Written informed consent was obtained from participants parents or guardians and, where possible, verbal assent was obtained from children.

Diagnosis and training

We conducted a one-week intensive interactive training led by two physicians with experience in the diagnosis and management of scabies, pyoderma and other common dermatoses in Africa. As part of this exercise two registered nurses were trained to undertake standardised skin examinations and diagnose common dermatoses. We utilised both classroom and supervised field training sessions on dermatology history and examination skills with an emphasis on common skin infections. Diagnostic criteria were based on algorithms that had been previously validated in both the Pacific and Africa[10,11]. Scabies was diagnosed on the presence of pruritus and typical skin lesions including papules, vesicles and pustules in a typical distribution[12]. Pyoderma was defined as any skin lesion with evidence of pus or crusts. Infected scabies was diagnosed when scabies was present with evidence of pyoderma in the same distribution. Tinea corporis was defined as patches with overlying scale. Tinea capitis was defined as areas of scalp hair loss with associated scale.

Study design, sampling and procedures

Two cross-sectional surveys of skin diseases were performed in the Bijagos Archipelago. The primary survey was conducted in February-March 2018 during the dry season. A probability proportionate to size methodology was used to select villages across the archipelago. In each selected village all children aged 1–9 were enrolled in the study. To assess for any evidence of seasonal variation we conducted a smaller secondary survey in the most populated island of the archipelago, Bubaque, during the wet season (June-July 2018). In the second survey we randomly selected 1 in 5 households in all 17 villages and again enrolled children in to the



survey. We collected demographic information for each individual enrolled in either survey and the trained nurses undertook a standardised examination for scabies, pyoderma and fungal infections. In the dry season survey we additionally identified individuals with chronic ulcers, possibly compatible with yaws, and performed a rapid diagnostic test for yaws[13]. Where a treatable dermatoses was identified we provided treatment in the community free of charge.

Data collection and statistical analysis

Data were collected on tablets using OpenDataKit. Data were analysed using R 3.4.2 (The R Foundation for Statistical Computing). We present count and prevalence data, with 95% confidence intervals, for each condition during both the dry season and wet season surveys. Adjusted odds ratios (AOR) for each skin dermatoses were calculated using a random-effects multivariable logistic regression model adjusting for demographic variables and clustering at the village level. Age was treated as a continuous variable. In a secondary analysis we compared the prevalence of each dermatoses between the dry season and wet season surveys. For the dry-season survey we conducted the survey alongside a trachoma survey which aimed to enrol approximately 1,000 children, whilst the wet-season survey was conducted as a stand-alone survey. We calculated that a sample size of 385 children would be powered to detect a prevalence of each of the skin conditions of 10% +/- 3%.

Results

Dry season survey

In the primary dry season survey we enrolled 1062 children of whom 508 (47.8%) were male with a median age of 5 years (IQR 3–7) across all the inhabited islands of the archipelago. Overall 268 had evidence of one skin infection (25.2%) and 50 had evidence of two or more skin infections (4.7%).

The most common diagnosis during the dry season was tinea capitis (154/1062, 14.5% - 95% CI 12.5–16.8%), followed by tinea corporis (84/1062, 7.9% - 95% CI 6.4–9.7%), pyoderma (82/1062, 7.7% - 95% CI 6.2–9.5%) and scabies (56/1062, 5.2% - 95%CI 4.0–6.8%). We identified 3 individuals (0.3%) with chronic ulcers but all had a negative RDT for yaws. Both tinea capitis (aOR 3.03, 95% CI 2.09–4.48) and tinea corporis (aOR 1.51, 95% CI 0.96–2.39) were more common in males although only the association with tinea capitis was statistically significant (p <0.01). There was no association between scabies and gender (aOR 1.29, 95% CI 0.74–2.25) nor age (aOR 0.89, 95%CI 0.81–1.0). Pyoderma was strongly associated with scabies (aOR 3.1, 95% CI 1.38–6.5). The population attributable risk fraction for pyoderma due to scabies was 7%. Risk factor data for each condition during the dry season survey is shown in Table 1.

Wet season survey

In the wet season survey we enrolled 320 children of whom 162 (50.6%) were male with a median age of 5 years (IQR 3–7). Overall 99 children had evidence of one skin infection (30.9%) and 22 children had evidence of two or more skin infections (6.9%).

During the wet season tinea capitis remained the most common diagnosis (79/320, 24.7% - 95% CI 20.1–29.9%), followed by pyoderma (38/320, 11.9% - 95% CI 8.6–16.1%), tinea corporis (23/320, 7.2% - 95% 4.7–10.7%) and scabies (6/320, 1.9% - 95% CI 0.8–4.2%). Male gender remained associated with the presence of tinea capitis (aOR 4, 95% CI 2.43–6.00). No variables were significantly associated with either tinea corporis or scabies in the wet season



Table 1. Association between demographic factors and common skin dermatoses during the dry season survey.

	Tinea Capitis		Tinea Corporis		Scabies		Pyoderma		
	Odds Ratio (95% CI)	Adjusted Odds Ratio (95% CI)							
Age (per year)	1.10 (1.03-1.18)*	1.09 (1.02–1.17)*	0.98 (0.89- 1.06)	0.97 (0.89–1.06)	0.9 (0.81- 1.0)	0.89 (0.81–1.0)	0.90 (0.82- 0.99)*	0.90 (0.82-0.99)*	
Male gender	3.03 (2.09- 4.48)**	2.97 (2.05-4.38)**	1.51 (0.96- 2.39)	1.53 (0.97-2.43)	1.24 (0.75– 2.15)	1.29 (0.74–2.25)	1.45 (0.91–2.3)	1.48 (0.93-2.38)	
Presence of Scabies	N/A	N/A	N/A	N/A	N/A	N/A	3.31 (1.47- 6.89)**	3.1 (1.38–6.5)**	

^{*} p < 0.05

https://doi.org/10.1371/journal.pntd.0007820.t001

survey. Increasing age was associated with a decreased risk of pyoderma (aOR per year of age 0.82, 95%CI 0.74–0.9). Scabies remained associated with pyoderma (aOR 3.0, 95% CI 0.04–15.04). The population attributable risk fraction for pyoderma due to scabies during the wet season survey was 3.5%. Risk factor data for each condition during the wet season survey is shown in Table 2. The prevalence of both tinea capitis (24.7% vs 14.5%, p <0.001) and pyoderma (11.9% vs 7.9%, p = 0.03) was significantly higher, whilst the prevalence of scabies was lower (1.9% vs 5.2%, p = 0.02) during the wet season survey compared to the dry season survey. The prevalence of each disease by age, gender and season are shown in Table 3.

Discussion

Our study shows the substantial burden of disease caused by common communicable dermatoses amongst children of the Bijagos Archipelago of Guinea-Bissau. In both the wet and dry season survey almost 30% of children had evidence of at least one communicable dermatosis highlighting the extent to which these easily treatable diseases are ubiquitous in this setting. A particular strength of this study is the robust, population-based cluster random sampling methodology that was used. The majority of similar studies to date have used less epidemiologically robust techniques which are at considerable risk of bias, possibly resulting in inaccurate prevalence estimates. By contrast our study used a gold-standard sampling methodology used in large-scale NTD mapping studies such as the Global Trachoma Mapping Project[14].

Table 2. Association between demographic factors and common skin dermatoses during the wet season survey.

	T	0 1				•			
	Tinea Capitis		Tinea Corpor	is	Scabies		Pyoderma		
	Odds Ratio (95% CI)	Adjusted Odds Ratio (95% CI)							
Age (per year)	0.95 (0.9–1.00)	0.95 (0.89–1.00)	1.07 (0.98- 1.15)	1.07 (0.98–1.16)	0.91 (0.74– 1.08)	0.91 (0.74.– 1.08)	0.82 (0.73- 0.89)**	0.82 (0.74-0.90)*	
Male gender	4.00 (2.43- 6.82)**	4.09 (2.48-6.98)**	0.60 (0.31– 1.15)	0.59 (0.03–1.13)	1.51 (0.37– 7.41)	1.56 (0.38–7.67)	0.98 (0.52– 1.88)	1.01 (0.52–1.96)	
Presence of Scabies	N/A	N/A	N/A	N/A	N/A	N/A	3.43 (0.48– 16.2)	3.00 (0.40–15.04)	

^{*} p < 0.05

https://doi.org/10.1371/journal.pntd.0007820.t002

^{**} p < 0.01

^{**} p < 0.01



Table 3. Prevalence of each skin infection by age, gender and season.

				Dry Season	1	Wet Season					
Age	Gender	Total	Tinea Capitis n (%)	Tinea Corporis n (%)	Scabies n (%)	Pyoderma n (%)	Total	Tinea Capitis n (%)	Tinea Corporis n (%)	Scabies n (%)	Pyoderma n (%)
0	Female	30	0 (0)	0 (0)	1 (3.4)	2 (6.7)	N/A	N/A	N/A	N/A	N/A
	Male	14	0 (0)	1 (7.2)	0 (0)	2 (14.3)	N/A	N/A	N/A	N/A	N/A
1	Female	44	3 (6.9)	6 (13.7)	4 (9.1)	6 (13.7)	12	0 (0)	0 (0)	0 (0)	2 (16.7)
	Male	38	5 (13.2)	4 (10.6)	2 (5.3)	4 (10.6)	14	1 (7.2)	1 (7.2)	2 (14.3)	2 (14.3)
2	Female	53	1 (1.9)	3 (5.7)	1 (1.9)	1 (1.9)	18	2 (11.2)	2 (11.2)	0 (0)	5 (27.8)
	Male	55	8 (14.6)	6 (11)	5 (9.1)	5 (9.1)	15	5 (33.4)	0 (0)	0 (0)	3 (20)
3	Female	76	7 (9.3)	5 (6.6)	6 (7.9)	9 (11.9)	15	3 (20)	0 (0)	0 (0)	2 (13.4)
	Male	54	12 (22.3)	10 (18.6)	5 (9.3)	6 (11.2)	19	5 (26.4)	0 (0)	0 (0)	1 (5.3)
4	Female	60	7 (11.7)	2 (3.4)	6 (10)	7 (11.7)	18	2 (11.2)	3 (16.7)	1 (5.6)	5 (27.8)
	Male	55	14 (25.5)	4 (7.3)	4 (7.3)	6 (11)	14	6 (42.9)	2 (14.3)	0 (0)	4 (28.6)
5	Female	64	4 (6.3)	8 (12.5)	2 (3.2)	2 (3.2)	24	3 (12.5)	1 (4.2)	1 (4.2)	2 (8.4)
	Male	48	12 (25)	2 (4.2)	2 (4.2)	8 (16.7)	27	7 (26)	2 (7.5)	1 (3.8)	1 (3.8)
6	Female	70	10 (14.3)	4 (5.8)	3 (4.3)	3 (4.3)	16	3 (18.8)	2 (12.5)	0 (0)	2 (12.5)
	Male	64	14 (21.9)	5 (7.9)	4 (6.3)	4 (6.3)	19	7 (36.9)	2 (10.6)	0 (0)	2 (10.6)
7	Female	53	7 (13.3)	0 (0)	3 (5.7)	4 (7.6)	20	3 (15)	2 (10)	0 (0)	0 (0)
	Male	57	17 (29.9)	6 (10.6)	3 (5.3)	2 (3.6)	17	5 (29.5)	0 (0)	0 (0)	2 (11.8)
8	Female	43	4 (9.4)	0 (0)	1 (2.4)	0 (0)	19	2 (10.6)	1 (5.3)	0(0)	1 (5.3)
	Male	63	16 (25.4)	6 (9.6)	3 (4.8)	4 (6.4)	26	16 (61.6)	2 (7.7)	0 (0)	2 (7.7)
9	Female	61	4 (6.6)	8 (13.2)	0 (0)	2 (3.3)	16	3 (18.8)	2 (12.5)	0 (0)	1 (6.3)
	Male	60	9 (15)	4 (6.7)	1 (1.7)	5 (8.4)	11	6 (54.6)	1 (9.1)	1 (9.1)	1 (9.1)

https://doi.org/10.1371/journal.pntd.0007820.t003

Fungal dermatoses were the most common dermatoses, in keeping with the findings of previous studies conducted elsewhere [1]. As previously reported [15], tinea capitis was more common amongst boys. The prevalence of scabies was relatively low compared to other studies in Africa [16]. This may, in part, be due to the round of MDA with ivermectin for the elimination of lymphatic filariasis carried out by the Ministry of Health in the previous year, which may have reduced the prevalence of scabies at this time [5,17–19]. There was a substantial burden of pyoderma in these communities and some evidence that this increased during the wet season. In other settings there is evidence that pyoderma is a driver of more serious bacterial infections, glomerulonephritis and rheumatic fever [20]. Although there is no specific data from Guinea-Bissau these sequelae including invasive *Staphylococcus aureus* and rheumatic fever are common in West Africa [21–23]. It is plausible that the burden of pyoderma in our current surveys may also be a driver for these conditions in Guinea-Bissau.

Our study has some limitations. Firstly, we examined only children aged nine years old or less. It is plausible that a different range of skin conditions are more common in older children and further work is required to explore this. Secondly, our study was limited to rural areas and should not be taken to be representative of peri-urban or urban populations in other regions of Guinea-Bissau. Our definition of tinea capitis was based on patches of hair loss. We might therefore have missed some cases of tinea capitis presenting with less common manifestations, such as kerion, but this is unlikely to significantly have affected our results. Whilst diagnoses in our study were made by a trained health worker and not by a dermatologist several studies have previously shown that a high degree of diagnostic accuracy can be achieved with training [10,11,24]; we are therefore confident that our clinical data are accurate.

WHO has proposed an integrated-approach to screening for NTDs involving the skin and has developed initial healthcare worker training materials to support this strategy [25]. Initial



small scale pilot studies have demonstrated clearly that these rarer, if more serious diseases, are dwarfed by the burden of disease of common skin conditions[7,8]. Whilst this might be considered a negative, from the perspective of identifying cases of NTDs, we argue that this is in fact a positive. First, common dermatoses represent differential diagnoses of rarer NTDs of the skin and training health care workers in their recognition is vital to ensure robust diagnoses. Secondly, by identifying more common skin problems integrated approaches can contribute to universal health coverage. Thirdly, focusing on the broader conditions affecting communities is likely to positively influence community acceptability of such screening strategies. Finally, given the burden caused by these common dermatoses, they deserve treatment in their own right as part of Universal Health Coverage to deliver 'skin-health for all' [26].

Supporting information

S1 Checklist. Strobe checklist. (DOCX)

S1 Data. Supporting dataset. (CSV)

Acknowledgments

We extend our thanks to colleagues in the Ministério de Saúde Pública in Bissau and to the communities of the Bijagós Archipelago, our field research team and study participants.

Author Contributions

Conceptualization: Michael Marks, Thomas Sammut, Adriana Goncalves, Anna Last, Stephen L. Walker.

Data curation: Michael Marks.

Formal analysis: Michael Marks, Thomas Sammut, Adriana Goncalves.

Funding acquisition: Anna Last.

Investigation: Thomas Sammut, Marito Gomes Cabral, Eunice Teixeira da Silva, Adriana Goncalves, Stephen L. Walker.

Methodology: Michael Marks, Thomas Sammut, Marito Gomes Cabral, Eunice Teixeira da Silva, Adriana Goncalves, Anna Last, Stephen L. Walker.

Project administration: Amabelia Rodrigues, Cristóvão Manjuba, Jose Nakutum, Janete Ca, Umberto D'Alessandro, Jane Achan, James Logan, Robin Bailey, Anna Last.

Resources: Amabelia Rodrigues, Cristóvão Manjuba, Jose Nakutum, Janete Ca, Umberto D'Alessandro, Jane Achan, James Logan, Robin Bailey.

Supervision: Michael Marks, Stephen L. Walker.

Writing - original draft: Michael Marks, Thomas Sammut.

Writing – review & editing: Marito Gomes Cabral, Adriana Goncalves, Amabelia Rodrigues, Cristóvão Manjuba, Jose Nakutum, Janete Ca, Umberto D'Alessandro, Jane Achan, James Logan, Robin Bailey, David Mabey, Anna Last, Stephen L. Walker.



References

- Hay RJ, Johns NE, Williams HC, Bolliger IW, Dellavalle RP, Margolis DJ, et al. The global burden of skin disease in 2010: an analysis of the prevalence and impact of skin conditions. J Invest Dermatol. 2014; 134: 1527–1534. https://doi.org/10.1038/jid.2013.446 PMID: 24166134
- Walker SL, Lebas E, Sario VD, Deyasso Z, Doni SN, Marks M, et al. The prevalence and association with health-related quality of life of tungiasis and scabies in schoolchildren in southern Ethiopia. PLoS Negl Trop Dis. 2017; 11: e0005808. https://doi.org/10.1371/journal.pntd.0005808 PMID: 28771469
- Romani L, Steer AC, Whitfeld MJ, Kaldor JM. Prevalence of scabies and impetigo worldwide: a systematic review. Lancet Infect Dis. 2015; 15: 960–967. https://doi.org/10.1016/S1473-3099(15)00132-2
 PMID: 26088526
- Krotneva SP, Coffeng LE, Noma M, Zouré HGM, Bakoné L, Amazigo UV, et al. African Program for Onchocerciasis Control 1995–2010: Impact of Annual Ivermectin Mass Treatment on Off-Target Infectious Diseases. PLoS Negl Trop Dis. 2015; 9: e0004051. https://doi.org/10.1371/journal.pntd.0004051 PMID: 26401658
- Martin D, Wiegand R, Goodhew B, Lammie P, Mkocha H, Kasubi M. Impact of Ivermectin Mass Drug Administration for Lymphatic Filariasis on Scabies in Eight Villages in Kongwa District, Tanzania. Am J Trop Med Hyg. 2018; https://doi.org/10.4269/ajtmh.18-0018 PMID: 30062986
- Enbiale W, Ayalew A. Investigation of a Scabies Outbreak in Drought-Affected Areas in Ethiopia. Trop Med Infect Dis. 2018; 3: 114. https://doi.org/10.3390/tropicalmed3040114 PMID: 30380650
- Yotsu RR, Kouadio K, Vagamon B, N'guessan K, Akpa AJ, Yao A, et al. Skin disease prevalence study in schoolchildren in rural Côte d'Ivoire: Implications for integration of neglected skin diseases (skin NTDs). PLoS Negl Trop Dis. 2018; 12: e0006489. https://doi.org/10.1371/journal.pntd.0006489 PMID: 29771976
- 8. Barogui YT, Diez G, Anagonou E, Johnson RC, Gomido IC, Amoukpo H, et al. Integrated approach in the control and management of skin neglected tropical diseases in Lalo, Benin. PLoS Negl Trop Dis. 2018; 12: e0006584. https://doi.org/10.1371/journal.pntd.0006584 PMID: 29939988
- Last AR, Burr SE, Weiss HA, Harding-Esch EM, Cassama E, Nabicassa M, et al. Risk Factors for Active Trachoma and Ocular Chlamydia trachomatis Infection in Treatment-Naïve Trachoma-Hyperendemic Communities of the Bijagós Archipelago, Guinea Bissau. PLoS Negl Trop Dis. 2014; 8: e2900. https://doi.org/10.1371/journal.pntd.0002900 PMID: 24967629
- Steer AC, Tikoduadua LV, Manalac EM, Colquhoun S, Carapetis JR, Maclennan C. Validation of an Integrated Management of Childhood Illness algorithm for managing common skin conditions in Fiji. Bull World Health Organ. 2009; 87: 173–179. https://doi.org/10.2471/BLT.08.052712 PMID: 19377712
- Mahé A, Faye O, N'Diaye HT, Ly F, Konaré H, Kéita S, et al. Definition of an algorithm for the management of common skin diseases at primary health care level in sub-Saharan Africa. Trans R Soc Trop Med Hyg. 2005; 99: 39–47. https://doi.org/10.1016/j.trstmh.2004.03.008 PMID: 15550260
- Engelman D, Fuller LC, Steer AC, Panel for the IA for the C of SD. Consensus criteria for the diagnosis of scabies: A Delphi study of international experts. PLoS Negl Trop Dis. 2018; 12: e0006549. https://doi.org/10.1371/journal.pntd.0006549 PMID: 29795566
- Marks M, Goncalves A, Vahi V, Sokana O, Puiahi E, Zhang Z, et al. Evaluation of a Rapid Diagnostic Test for Yaws Infection in a Community Surveillance Setting. PLoS Negl Trop Dis. 2014; 8: e3156. https://doi.org/10.1371/journal.pntd.0003156 PMID: 25211018
- 14. Solomon AW, Pavluck AL, Courtright P, Aboe A, Adamu L, Alemayehu W, et al. The Global Trachoma Mapping Project: Methodology of a 34-Country Population-Based Study. Ophthalmic Epidemiol. 2015; 22: 214–225. https://doi.org/10.3109/09286586.2015.1037401 PMID: 26158580
- Coulibaly O, L'Ollivier C, Piarroux R, Ranque S. Epidemiology of human dermatophytoses in Africa. Med Mycol. 2018; 56: 145–161. https://doi.org/10.1093/mmy/myx048 PMID: 28992062
- 16. Whittle HC, Abdullahi MT, Fakunle F, Parry EHO, Rajkovic AD. Scabies, pyoderma and nephritis in Zaria, Nigeria. A clinical and epidemiological study. Trans R Soc Trop Med Hyg. 1973; 67: 349–363. https://doi.org/10.1016/0035-9203(73)90112-0 PMID: 4591502
- Romani L, Marks M, Sokana O, Nasi T, Kamoriki B, Cordell B, et al. Efficacy of mass drug administration with ivermectin for control of scabies and impetigo, with coadministration of azithromycin: a single-arm community intervention trial. Lancet Infect Dis. 2019;0. https://doi.org/10.1016/S1473-3099(18) 30790-4 PMID: 30956111
- 18. Marks M, Toloka H, Baker C, Kositz C, Asugeni J, Puiahi E, et al. Randomized Trial of Community Treatment With Azithromycin and Ivermectin Mass Drug Administration for Control of Scabies and Impetigo. Clin Infect Dis. 2019; 68: 927–933. https://doi.org/10.1093/cid/ciy574 PMID: 29985978





- Romani L, Whitfeld MJ, Koroivueta J, Kama M, Wand H, Tikoduadua L, et al. Mass Drug Administration for Scabies Control in a Population with Endemic Disease. N Engl J Med. 2015; 373: 2305–2313. https://doi.org/10.1056/NEJMoa1500987 PMID: 26650152
- Parks T, Smeesters PR, Steer AC. Streptococcal skin infection and rheumatic heart disease. [Miscellaneous Article]. Curr Opin Infect Dis April 2012. 2012; 25: 145–153. https://doi.org/10.1097/QCO.0b013e3283511d27 PMID: 22327467
- Al AO et. Staphylococcus aureus Bacteremia in Children of Rural Areas of The Gambia, 2008–2015— Volume 25, Number 4—April 2019—Emerging Infectious Diseases journal—CDC. https://doi.org/10. 3201/eid2504.180935
- 22. Darboe S, Okomo U, Muhammad A-K, Ceesay B, Jallow M, Usuf E, et al. Community-acquired Invasive Bacterial Disease in Urban Gambia, 2005–2015: A Hospital-based Surveillance. Clin Infect Dis. 2019; 69: S105–S113. https://doi.org/10.1093/cid/ciz463 PMID: 31505627
- 23. Watkins DA, Johnson CO, Colquhoun SM, Karthikeyan G, Beaton A, Bukhman G, et al. Global, Regional, and National Burden of Rheumatic Heart Disease, 1990–2015. N Engl J Med. 2017; 377: 713–722. https://doi.org/10.1056/NEJMoa1603693 PMID: 28834488
- 24. Mahé A, Faye O, N'Diaye HT, Konaré HD, Coulibaly I, Kéita S, et al. Integration of basic dermatological care into primary health care services in Mali. Bull World Health Organ. 2005; 7.
- 25. World Health Organization. Recognizing neglected tropical diseases through changes on the skin [Internet]. World Health Organization; 2018 Jun. Available: http://www.who.int/neglected_diseases/resources/9789241513531/en/
- 26. What We Do. In: ILDS [Internet]. [cited 26 May 2019]. Available: https://ilds.org/what-we-do/